Author's response to reviews

Title: Effectiveness and cost-effectiveness of a telehealth intervention to support the management of long-term conditions: study protocol for two linked randomized controlled trials

Authors:

Clare L Thomas (clare.thomas@bristol.ac.uk)
Mei-See Man (Mei-see.Man@bristol.ac.uk)
Alicia O'Cathain (a.ocathain@sheffield.ac.uk)
Sandra Hollinghurst (s.p.hollinghurst@bristol.ac.uk)
Shirley Large (shirley.large@nhsdirect.nhs.uk)
Louisa Edwards (Louisa.Edwards@bristol.ac.uk)
Jon Nicholl (j.nicholl@sheffield.ac.uk)
Alan A Montgomery (alan.montgomery@nottingham.ac.uk)
Chris Salisbury (c.salisbury@bristol.ac.uk)

Version: 2
Date: 19 December 2013

Author's response to reviews:

Please see the attached cover letter describing revisions to the manuscript